Author’s response to reviews

Title: Limited efficacy of adalimumab in the acute phase of serpiginous choroiditis refractory to corticosteroid and cyclosporine, A case report

Authors:
Kazunori Noda (nodakazu@kuhp.kyoto-u.ac.jp)
Akio Oishi (aquio@kuhp.kyoto-u.ac.jp)
Akihito Uji (akihito1@kuhp.kyoto-u.ac.jp)
Saori Tanaka (ds110950@kuhp.kyoto-u.ac.jp)
Akitaka Tsujikawa (tujikawa@kuhp.kyoto-u.ac.jp)

Version: 2 Date: 04 Apr 2019

Author’s response to reviews:

We appreciate your consideration on our manuscript.

We have incorporated the comments.

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are some issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

   Please note that use of a professional language service is not a guarantee of acceptance for publication.

   > We had the manuscript edited by a professional English editing service.

2. In the main text of the manuscript, please change the heading "CASE REPORT" to "Case presentation".

   > The heading was revised.

3. Please change the heading "Discussion" to "Discussion and conclusions".

   > The heading was revised.
4. In the section 'Funding', please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

> The statement was added.

5. Please remove the CARE checklist and the cover letter from the file inventory as this is no longer needed at this time.

> CARE checklist and the cover letter were removed from the submission.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

> We uploaded clean version of the manuscript.